

REVIEW HISTORY

Manuscript Title: *Interpretation of Myth in Jennifer Saint's Novel Ariadne*

Submitted: 06 May, 2025

Accepted: 09 June, 2025

Published: 12 June, 2025

Corresponding Author: **Nana Mukeria**

Email: n.mukeria@europeanschool.ge

Doi: <http://dx.doi.org/10.19044/llc.v12no1a5>

PEER REVIEW

Reviewer 1: Olivera Petrović

Reviewer 2: Wafaa Abdel-Kader Mostafa

Reviewer 3: Ekaterine Kobakhidze